Author's response to reviews

Title: Charcot osteoarthropathy in type 2 diabetes persons presenting to specialist diabetes clinic at a tertiary care hospital.

Authors:

Bilal Bin Younis (bilalbin@gmail.com)
Adeela Shahid (dradeelas@hotmail.com)
Rozina Arshad (rozina979@gmail.com)
Saima Khurshid (dr.cyma@yahoo.com)
Junaid Masood (Dph_junaidmasood@hotmail.com)

Version: 4
Date: 5 May 2015

Author's response to reviews: see over
Based upon Reviewer’s comments

1. Statistical method of analysis has been changed. We have used the Kruskal Wallis test instead of Pearson collection and it has been updated in the revised manuscript under heading Material and Methods, sub heading 2.7 Statistical analysis, line no 150.

2. The Table 3 has been updated with Kruskal Wallis test results on page on 17 of the revised manuscript.

3. The Results of the statistical analysis has been added under heading Results, sub heading 3.2 Charcot osteoarthropathy line no 165-169.

4. The manuscript has been edited for possible grammatical errors